Author’s response to reviews

Title: Planning for globally coordinated cessation of bivalent oral poliovirus vaccine: Risks of non-synchronous cessation and unauthorized oral poliovirus vaccine use

Authors:

Radboud Duintjer Tebbens (rdt@kidrisk.org)

Lee M Hampton (euf0@cdc.gov)

Kimberly M Thompson (kimt@kidrisk.org)

Version: 2 Date: 19 Mar 2018

Author’s response to reviews:

Dear Dr. Devoto, Thank you for your recommendation and for handling our manuscript. Per your request, here is a clean version of the manuscript without tracked changes or highlighting and with a more detailed statement about the role of the funding body, which reads: "The Centers for Disease Control and Prevention (CDC) supported the work of RJDT and KMT under Cooperative Agreement Number 5NU2RGH001913-02-00. The contents of this manuscript are solely the responsibility of the authors and do not necessarily represent the official views of the CDC. The CDC did not participate in the design of the study or the collection, analysis, or interpretation of data other than through the authorship of LHM.". We look forward to receiving your final decision as soon as possible. Best, Radboud Duintjer Tebbens